Author's response to reviews

Title: Determinants of serum concentrations of organochlorine compounds in Swedish pregnant women: a cross-sectional study

Authors:

Anders Glynn (anders.glynn@slv.se)
Marie Aune (marie.aune@slv.se)
Per Ola Darnerud (poda@slv.se)
Sven Cnattingius (sven.cnattingius@mep.ki.se)
Rickard Bjerselius (ribj@slv.se)
Wulf Becker (wulf.becker@slv.se)
Sanna Lignell (sanna.lignell@slv.se)

Version: Date: 30 January 2007

Author's response to reviews:

Dear Editors,

Here is the revised manuscript 2026881856114056. The following changes have been made after the editorial comments:

1. Please ensure the title in the submission system and that of the manuscript match. Done.

2. Keywords section - Please remove the keywords section from the manuscript. Done.

3. Tables - Tables must be divided into cells / fields. The borders of tables included as part of the manuscript file must be visible black lines. Tables generated with tabbed text are not acceptable. Please ensure that text alignment in columns is clear and consistent. See Table 1, the percent figures may be pushed to the top of the cell if the table hasn't been properly divided in to cells. Table 1 has been revised. Table text has been aligned.

4. Therefore, when you submit your final version, please carefully check your manuscript for any typing or content errors that might have inadvertently occurred, and for small changes to the text that you or your co-authors wish to make at this time. We have checked the ms. The language in the text has been revised a little bit.

Best whishes

Anders